Author's response to reviews

Title: A multi-centre randomised controlled study of pre-IVF outpatient hysteroscopy in women with recurrent IVF implantation failure: Trial of Outpatient Hysteroscopy - [TROPHY] in IVF

Authors:

Tarek El-Toukhy Dr (tarek.el-toukhy@gstt.nhs.uk)
Campo Rudi Mr (rudi.campo@lifeleuven.be)
Sesh K Sunkara Dr (sesh.sunkara@kcl.ac.uk)
Yakoub Khalaf Dr (yakoub.khalaf@kcl.ac.uk)
Arri Coomarasamy Dr (a.coomarasamy@bham.ac.uk)

Version: 3 Date: 27 November 2009

Author's response to reviews:

Submitting Formatting Changes

To The BioMed Central Editorial Production Team,

We have made the formatting changes to the final version of the manuscript as per your request.

Author Affiliations have been added to the manuscript.

The text box included in the manuscript has been deleted.

References 24 - 26 are included in the manuscript. Please find ref 24 on page 6, ref 25 on page 7 and ref 26 on page 10.

A functioning weblink has been provided to reference 2.

Figure citation (referring to figure 2) on page 10 has been deleted.

A section Additional files has been added to the end of the manuscript, with the title and details of the manuscript.

Author qualifications have been deleted.

All highlighting has been removed from the text.
All abbreviations have been provided as a paragraph.

Thanking you

Yours sincerely

Dr Tarek El-Toukhy

(On behalf of the authors)